Author's response to reviews

Title: Views on sick-listing practice among Swedish General Practitioners - a phenomenographic study

Authors:

Malin S Swartling (malin.swartling@rehab.uu.se)
Stefan A Peterson (stefan.peterson@ki.se)
Rolf A Wahlstrom (rolf.wahlstrom@ki.se)

Version: 2 Date: 29 March 2007

Author's response to reviews:

Dear Editor,

We have checked a printout of the manuscript. Unfortunately the tables do not come out nicely. We therefore intend to insert page breaks for each table.

Kind regards

Malin Swartling